Author's response to reviews

Title: Rapid improvement in verbal fluency and aphasia following perispinal etanercept in Alzheimer's disease

Authors:

Edward L. Tobinick (etmd@ucla.edu)
Hyman Gross (hymangross@yahoo.com)

Version: 9 Date: 7 July 2008

Author's response to reviews:

Sentence confirming permission to use video added in methods section as requested by editors

One reference cite corrected (2nd 46 changed to 47)

Three typos corrected (number of patients in clinical trial who were tested with verbal fluency measures was 15-3=12); this was correct in the abstract but required correction at two places in the text (pages 19 and 24) and in the caption for figure 1.